

Citation: Conrad N, Judge A, Canoy D, Tran J, O'Donnell J, Nazarzadeh M, et al. (2019) Diagnostic tests, drug prescriptions, and follow-up patterns after incident heart failure: A cohort study of 93,000 UK patients. PLoS Med 16(5): e1002805. https://doi.org/10.1371/journal.pmed.1002805

Academic Editor: Aziz Sheikh, Edinburgh University, UNITED KINGDOM

Received: November 19, 2018

Accepted: April 12, 2019

Published: May 21, 2019

Copyright: © 2019 Conrad et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Data cannot be shared publicly and is subject to a licence agreement. Researchers who wish to obtain access to data should contact the CPRD's Independent Scientific Advisory Committee (ISAC) (contact via isac@cprd.com).

Funding: NC is supported by the British Heart Foundation. KR, DC, and FDRH are supported by the National Institute of Health Research (NIHR) Oxford Biomedical Research Centre. KR further receives grants from the Oxford Martin School, as **RESEARCH ARTICLE**

Diagnostic tests, drug prescriptions, and follow-up patterns after incident heart failure: A cohort study of 93,000 UK patients

Nathalie Conrad¹, Andrew Judge^{2,3,4}, Dexter Canoy^{1,5,6}, Jenny Tran^{1,6}, Johanna O'Donnell¹, Milad Nazarzadeh^{1,6,7}, Gholamreza Salimi-Khorshidi¹, F. D. Richard Hobbs⁸, John G. Cleland⁹, John J. V. McMurray¹⁰, Kazem Rahimi^{1,5,6,11}*

1 The George Institute for Global Health, University of Oxford, Oxford, United Kingdom, 2 Nuffield Department of Orthopaedics, Rheumatology and Musculoskeletal Sciences, Nuffield Orthopaedic Centre, University of Oxford, Oxford, United Kingdom, 3 Bristol National Institute for Health Research Biomedical Research Centre, Musculoskeletal Research Unit, University of Bristol, Southmead Hospital, Bristol, United Kingdom, 4 Medical Research Council Lifecourse Epidemiology Unit, University of Southampton, Southampton General Hospital, Southampton, United Kingdom, 5 National Institute for Health Research Oxford Biomedical Research Centre, University of Oxford, Oxford, United Kingdom, 6 Deep Medicine, Oxford Martin School, University of Oxford, Oxford, United Kingdom, 7 Collaboration Center of Meta-Analysis Research, Torbat Heydariyeh University of Medical Sciences, Torbat Heydariyeh, Iran, 8 Nuffield Department of Primary Care Health Sciences, Radcliffe Observatory Quarter, University of Oxford, Oxford, United Kingdom, 9 Robertson Centre for Biostatistics and Clinical Trials, University of Glasgow and National Heart & Lung Institute, Imperial College London, London, United Kingdom, 10 Institute of Cardiovascular and Medical Sciences, University of Glasgow, Glasgow, United Kingdom, 11 Oxford University Hospitals National Health Service Foundation Trust, Oxford, United Kingdom

* kazem.rahimi@georgeinstitute.ox.ac.uk

Abstract

Background

Effective management of heart failure is complex, and ensuring evidence-based practice presents a major challenge to health services worldwide. Over the past decade, the United Kingdom introduced a series of national initiatives to improve evidence-based heart failure management, including a landmark pay-for-performance scheme in primary care and a national audit in secondary care started in 2004 and 2007, respectively. Quality improvement efforts have been evaluated within individual clinical settings, but patterns of care across its continuum, although a critical component of chronic disease management, have not been studied. We have designed this study to investigate patients' trajectories of care around the time of diagnosis and their variation over time by age, sex, and socioeconomic status.

Methods and findings

For this retrospective population-based study, we used linked primary and secondary health records from a representative sample of the UK population provided by the Clinical Practice Research Datalink (CPRD). We identified 93,074 individuals newly diagnosed with heart failure between 2002 and 2014, with a mean age of 76.7 years and of which 49% were women. We examined five indicators of care: (i) diagnosis care setting (inpatient or

well as the PEAK Urban programme from the UKRI's Global Challenge Research Fund - Grant Ref: ES/P011055/1. AJ is supported by the NIHR Biomedical Research Centre at the University Hospitals Bristol NHS Foundation Trust and the University of Bristol. FDRH further acknowledges support from the NIHR School for Primary Care Research (SPCR), and the NIHR Collaboration for Leadership in Applied Research in Health and Care (CLARHC) Oxford. JOD acknowledges the support of the RCUK Digital Economy Programme. The funders of the study had no role in study design, data collection, data analysis, data interpretation, or writing of the report.

Competing interests: I have read the journal's policy and the authors of this manuscript have the following competing interests: NC's work is funded by a research grant from the British Heart Foundation. AJ has received consultancy fees from Freshfields Bruckhaus Deringer and has held advisory board positions (which involved receipt of fees) from Anthera Pharmaceuticals, Inc., outside of the submitted work. JT declared funding for scholarships and grants from Rhodes Trust, Clarendon Fund, and British Research Council. JGFC has received honoraria or research support from Amgen, Bayer, Medtronic, Novartis, and Servier, FDRH has received occasional funding from Novartis in relation to speaking or consultancy on heart failure in the past 3 years. KR receives a stipend as a Speciality Consulting Editor for PLOS Medicine and serves on the journal's Editorial Board.

Abbreviations: ACE-I, angiotensin-convertingenzyme inhibitor; ARB, angiotensin receptor blocker; BMI, body-mass index; BNP, B-type natriuretic peptide; CI, confidence interval; CPRD, Clinical Practice Research Datalink; ECG, electrocardiogram; ESC, European Society of Cardiology; HES, Hospital Episodes Statistics; IMD, Index of Multiple Deprivation; ISAC, Independent Scientific Advisory Committee; MRA, mineralocorticoid receptor antagonist; NA, not applicable; NHFA, National Heart Failure Audit; NP, natriuretic peptide; NT-pro-BNP, N-terminal-pro-BNP; QOF, Quality and Outcomes Framework; RECORD, REporting of studies Conducted using Observational Routinely collected health Data; RR, risk ratio.

outpatient), (ii) posthospitalisation follow-up in primary care, (iii) diagnostic investigations, (iv) prescription of essential drugs, and (v) drug treatment dose. We used Poisson and linear regression models to calculate category-specific risk ratios (RRs) or adjusted differences and 95% confidence intervals (CIs), adjusting for year of diagnosis, age, sex, region, and socioeconomic status. From 2002 to 2014, indicators of care presented diverging trends. Outpatient diagnoses and follow-up after hospital discharge in primary care declined substantially (ranging from 56% in 2002 to 36% in 2014, RR 0.64 [0.62, 0.67] and 20% to 14%, RR 0.73 [0.65, 0.82], respectively). Primary care referral for diagnostic investigations and appropriate initiation of beta blockers and angiotensin-converting-enzyme inhibitors (ACE-Is) or angiotensin receptor blockers (ARBs) both increased significantly (37% versus 82%, RR 2.24 [2.15, 2.34] and 18% versus 63%, RR 3.48 [2.72, 4.43], respectively). Yet, the average daily dose prescribed remained below guideline recommendations (42% for ACE-Is or ARBs, 29% for beta blockers in 2014) and was largely unchanged beyond the first 30 days after diagnosis. Despite increasing rates of treatment initiation, the overall dose prescribed to patients in the 12 months following diagnosis improved little over the period of study (adjusted difference for the combined dose of beta blocker and ACE-I or ARB: +6% [+2%, +10%]). Women and patients aged over 75 years presented significant gaps across all five indicators of care. Our study was limited by the available clinical information, which did not include exact left ventricular ejection fraction values, investigations performed during hospital admissions, or information about follow-up in community heart failure clinics.

Conclusions

Management of heart failure patients in the UK presents important shortcomings that affect screening, continuity of care, and medication titration and disproportionally impact women and older people. National reporting and incentive schemes confined to individual clinical settings have been insufficient to identify these gaps and address patients' long-term care needs.

Author summary

Why was this study done?

- Heart failure is a common, costly, and severe condition—it affects about 2% of the population in high-income countries, with mortality rates comparable to the gravest forms of cancer.
- Effective treatments exist but involve a complex process of investigations, stepwise initiation of medicines, and dose adjustments that in practice is often challenging to follow.
- Over the past decade, the UK introduced separate programmes to evaluate and improve the management of heart failure patients in primary and secondary care. However, patients' trajectories of care across different healthcare settings have not been studied.

What did the researchers do and find?

- We aimed to investigate the medical care received by heart failure patients in the UK from diagnosis up to a year later.
- We used electronic health records from 93,000 patients who were diagnosed with heart failure between 2002 and 2014. We investigated trajectories of care, including where patients were diagnosed, investigations that were performed, and medications they received, examined changes over time and if results differed according to patients' age, sex, or socioeconomic status.
- Our results show that the medical care received by heart failure patients presents important gaps. Patients were more likely to be diagnosed in hospital rather than by their general practitioner, received insufficient follow-up after hospitalisation, and dosages prescribed to patients were too low. Moreover, gaps in care were more common in women, older people, and, to some extent, socioeconomically deprived individuals.

What do these findings mean?

- Heart failure care in the UK presents important gaps that pertain to patients' long-term care needs and are likely to negatively impact patients' health and quality of life.
- Nationwide quality improvement programmes have failed to identify these gaps, essentially because they evaluate each individual healthcare setting on its own. To further improve patient care, health systems may need to reflect on patients' journey across the continuum of care.
- Women, older people, and deprived populations are particularly prone to receiving suboptimal care. Research that helps us understand the reasons behind these disparities, or models of care that take each group's specificities into account, could help.

Introduction

Over the past 25 years, we have witnessed remarkable developments in clinical interventions that improve symptoms, quality of life, and prognosis in patients with heart failure. However, effective clinical care involves a complex process of investigations, stepwise initiation of medicines, and dose titration that often takes place in different care settings over several months and can be difficult to implement consistently. Although clinical guidelines provide a valuable tool to support physicians in the management of heart failure patients [1-7], ensuring optimal use of evidence-based therapies in routine clinical practice remains a major concern and challenge to health services worldwide [8,9].

In the UK, two major programmes have been introduced to improve physicians' adherence to evidence-based practices: the 'Quality and Outcomes Framework' (QOF) [10], a healthcare reporting and incentive scheme that applies to primary care physicians and was initiated in 2004; and the 'National Heart Failure Audit' (NHFA) [11], a large-scale reporting scheme for secondary care launched in 2007. Each scheme individually produces yearly reports on selected heart failure care indicators with very high rates of adherence to clinical guidelines

[12,13]. Yet, programmes only consider the clinical setting for which they were designed and have not assessed their broader impact on patient care across the continuum of primary and secondary services, which chronic conditions, such as heart failure, rely on. More generally, similar limitations apply to studies that have investigated heart failure care in Western countries. So far, studies have been largely confined to specific clinical settings, selected cohorts, and limited follow-up information, with no ability to describe and compare patient trajectories following a new diagnosis of heart failure (S1 Table). Finally, little is known of variations in care practices by important patient characteristics, such as age or sex.

To address these knowledge gaps, we used a database of linked primary and secondary healthcare records in the UK [14] and performed a longitudinal assessment of outpatient care, covering diagnosis, follow-up, diagnostic investigations, treatment initiation, and dosages in patients with incident heart failure. We further sought to investigate temporal trends and variation by important patient characteristics such as age, sex, and socioeconomic status.

Methods

Data source

We used electronic health records from the Clinical Practice Research Datalink (CPRD) from 1 January 1985 to 30 September 2015. The CPRD database contains anonymised patient data from approximately 7% of the current UK population and is broadly representative in terms of age and sex. CPRD is one of the largest databases of longitudinal medical records from primary care in the world and has been validated for epidemiological research for a broad range of conditions [14]. Primary care records include demographic information, consultations, drug prescriptions, diagnostic investigations, and referrals to specialists and were linked to secondary care admission records from Hospital Episodes Statistics (HES), which provides information on hospital admissions and related discharge diagnoses. Scientific approval for this study was given by the CPRD Independent Scientific Advisory Committee (ISAC).

Study population

Patients were men and women aged 16 and over, with records labelled as 'acceptable' for research purposes by CPRD quality control [14], approved for CPRD and HES linkage, and registered with their general practice for at least 12 months. Amongst these 4.0 million patients, we identified 93,074 with incident heart failure during the period 1 January 2002 to 31 December 2014. Incident heart failure diagnoses were defined as the first record of heart failure in primary care or hospital admission recorded in any diagnostic position and were identified following previously published methods [15].

Study outcomes

We examined five aspects of care across the disease trajectory of patients with heart failure: (i) diagnosis care setting, (ii) posthospitalisation follow-up, (iii) diagnostic investigations, (iv) prescription of essential medicines, and (v) treatment dosages. Data about diagnostic investigations and prescriptions were only available in primary care records, so indicators (iii), (iv), and (v) were restricted to patients whose heart failure was recorded in primary care (n = 47,925). Analysis of guideline-recommended drug treatment was further restricted to patients with reduced ejection fraction (n = 11,040) with no record of drug-class–specific contraindications or intolerance (S2 and S3 Tables). Fig 1 provides an overview of study outcome measures, with further details provided below.

			I.		Denominator
Diagnosis care setting	Care setting in which heart failure was first diagnosed (inpatient or outpatient).				All patients newly diagnosed with heart failure (n=93,074)
Post-hospitalisation follow-up	Heart failure record in primary care within 12 months of incident diagnosis in hospital.			+12 months	Patients diagnosed in hospital and discharged alive (n= 40,856)
Diagnostic investigations	Investigations (echocardiogram, ECG, NP test or specialist assessment) referred for within ± 6 months of incident heart failure.	, -6 months	+6 months		Primary care patients (n=47,946) ¹
Treatment initiation	Issue of at least 1 prescription within 3 months incident heart failure, of either ACE-I/ARB; beta blocker; MRA; or ACE-I/ARB and beta-blocker	of a-	+3 months		Primary care patients with reduced ejection fraction and no drug- specific contraindication
Treatment dose	Average daily dose prescribed in the 12 month following incident heart failure, as % of guidelin indicated target dose, of either ACE-I/ARB; bei blocker; MRA; or ACE-I/ARB and beta-blocker	s e- :a- :3		+12 months	or intolerance ²
		Heart f diagr	failure nosis		

¹Primary care patients include all outpatient diagnoses and inpatients with post-hospitalisation follow-up. ²For ACE-I/ARB n=9,920; for beta-blocker n=8,430; for MRA n=10,896; for ACE-I/ARB and beta-blocker n=7,768. ³In sub-group analyses we further investigated the average daily dose prescribed in the 3 months preceding diagnosis.

Fig 1. Definitions of essential care indicators. ACE-I, angiotensin-converting-enzyme inhibitor; ARB, angiotensin receptor blocker; ECG, electrocardiogram; MRA, mineralocorticoid receptor antagonist; NP, natriuretic peptide.

https://doi.org/10.1371/journal.pmed.1002805.g001

The care setting in which heart failure was first diagnosed was categorised as either inpatient or outpatient. Inpatient diagnoses were further categorised based on whether heart failure was listed in primary or secondary diagnostic position. Outpatient diagnoses refer to diagnoses first recorded in primary care with no prior hospitalisation and are likely to reflect both outpatient consultations by specialists and direct diagnoses by general practitioners.

The potential for primary care follow-up after diagnosis in hospital (in short follow-up) was defined as the documentation of heart failure in primary care records within 12 months of diagnosis, with the rationale that documentation of diagnosis is a prerequisite for subsequent disease monitoring and management (S1 Text).

Diagnostic investigations included tests recommended as "essential" by the European Society of Cardiology (ESC) guidelines [6]: (i) echocardiogram, (ii) electrocardiogram (ECG), and (iii) plasma natriuretic peptides (B-type natriuretic peptide [BNP] or N-terminal-pro-BNP [NT-pro-BNP]), as well as (iv) cardiology specialist assessment, within ± 6 months of incident heart failure diagnosis. Diagnostic tests were considered individually and as a composite of any of the four tests. The list of diagnostic codes used to identify diagnostic tests is presented in S4 Table.

Drug treatment patterns were investigated for the three main treatment classes indicated in the management of heart failure with reduced ejection fraction by the ESC guidelines [6]: (i) angiotensin-converting-enzyme inhibitors (ACE-Is) or angiotensin receptor blockers (ARBs), (ii) beta blockers, and (iii) mineralocorticoid receptor antagonists (MRAs).

For each of the three drug classes, we report treatment initiation as the proportion of eligible patients who received at least one prescription in the first three months following their heart failure diagnosis. We defined a composite treatment initiation indicator as prescriptions for the two treatment classes indicated in all patients with heart failure with reduced ejection fraction, which are ACE-Is or ARBs and beta blockers.

We further calculated the average daily dose prescribed to eligible patients over all days alive and registered with their general practice in the first 12 months following incident heart failure (S2 Text). We present average daily doses as a fraction of the drug-specific guideline-recommended dose. Daily doses were calculated for all eligible patients, irrespective of receipt of treatment, as an estimate of population level of treatment. The guideline-recommended doses were defined as the minimal target dose recommended by the latest ESC guidelines available during the study period (S5 Table). Average daily dose was analysed as a continuous variable, as a binary variable (<50% or \geq 50% of recommended dose), as well as categorised into 0%, 1%–24%, 25%–49%, 50%–74%, and \geq 75% of the recommended dose. Because drugs used for treatment of heart failure are also used for other indications, we further investigated average daily dose prescribed in the 3 months preceding diagnosis.

Patient characteristics

We extracted the most recent measurement of baseline characteristics within two years preceding incident heart failure diagnosis including systolic and diastolic blood pressure, smoking status, and body-mass index (BMI). BMI was categorised as underweight (<18.5 kg/m²), normal (18.5–24.9 kg/m²), overweight (25–29.9 kg/m²), and obese (\geq 30 kg/m²).

We further extracted information on comorbidities and socioeconomic status. To describe comorbidities, we selected 17 common chronic conditions (anaemia, asthma, atrial fibrillation, cancer, chronic kidney disease, chronic obstructive pulmonary disease, dementia, depression, diabetes, dyslipidaemia, hypertension, ischaemic heart disease, obesity, osteoarthritis, peripheral arterial disease, stroke, and thyroid disease). Diagnosis code lists for the extraction of each condition were adapted from the CALIBER code repository [16]. To describe socioeconomic status, we used patients' Index of Multiple Deprivation (IMD) 2015 quintile [17], a composite measure of relative deprivation at a small area level, ranked in ascending order of deprivation score and grouped in equal fifths.

Statistical analyses

For binary or categorical outcome variables, we computed age, sex, socioeconomic, and yearspecific risks as the proportion of eligible patients who received care within a defined timeframe of incident heart failure diagnosis. To assist readability, we further refer to risks, proportions, and rates interchangeably. For continuous outcome variables, we computed means and standard deviations.

To examine changes over time and by subgroups, we used Poisson or linear regression models with robust error variance and report risk ratios (RRs) or adjusted mean differences alongside corresponding 95% confidence intervals (CIs). All models were adjusted for year of diagnosis, age (categorised as <45, 45–54, 55–64, 65–74, and \geq 75 years), sex, region, and socioeconomic status. As a measure of temporal trend, we further report *p*-values modelled by including year as a continuous variable. Selected graphical representations were smoothed using local polynomial regression and labelled as such in the figure caption [18,19].

We performed several sensitivity analyses to assess the robustness of our results towards differences in follow-up duration, 30-day mortality, or prevalence of comorbidities over time and by subgroups (<u>S3 Text</u>).

Study findings are reported in accordance with the REporting of studies Conducted using Observational Routinely collected health Data (RECORD) recommendations (<u>S4 Text</u>) [20].

Analyses were prospectively specified in the study protocol (<u>\$5 Text</u>), and all statistical analyses were performed in R, version 3.4.2 (R Foundation for Statistical Computing, Vienna, Austria).

Results

A total of 93,074 patients newly diagnosed with heart failure between 2002 and 2014 were included in the study. Patient characteristics stratified by sex, socioeconomic status, and time period categories have been previously published [15]. Patient characteristics varied depending on where they were first diagnosed; those diagnosed in hospital were older, had more comorbidities, and were more likely to be women (Table 1). Patients with a record of reduced ejection fraction were more likely to be younger, be male, and present with fewer comorbidities than those for whom ejection fraction was preserved or unspecified (S6 Table).

Temporal trends of essential care indicators

Setting of heart failure diagnosis. Between 2002 and 2014, a diagnosis of heart failure was first recorded during a hospitalisation in 56% of cases (12% as a primary diagnosis and 44% as a secondary diagnosis) and in an outpatient setting in 44% (Table 2). Amongst secondary inpatient diagnoses of heart failure, about a third of primary causes for admission were cardiac and a fifth were respiratory disease.

Whilst the proportion of patients diagnosed in hospital with primary heart failure diagnosis remained stable at 12%, there were opposing trends for secondary inpatient diagnoses versus outpatient diagnoses. Rates of outpatient diagnoses showed a 36% relative decline over time (from 56% in 2002 to 36% in 2014, RR 0.64 [0.62, 0.67]) and were offset by a 67% relative increase in secondary inpatient diagnoses (from 31% in 2002 to 52% in 2014, RR 1.67 [1.60, 1.74]) (Table 2).

Primary care follow-up after hospitalisation. Amongst heart failure inpatients who survived the index hospitalisation, only 17% had their heart failure diagnosis recorded by their primary care physician in the subsequent 12 months, with rates declining over time (20% in 2002 versus 14% in 2014; RR 0.73 [0.65, 0.82]) (Fig 2). Follow-up rates were higher for patients with a primary discharge diagnosis of heart failure (31%) than secondary diagnosis (12%) (Table 2).

Diagnostic tests. Amongst patients with a recorded diagnosis of heart failure in primary care, the use of diagnostic investigations increased substantially over time. In 2014, 82% of patients had at least one of the following tests: NP (8%), an ECG (37%), an echocardiogram (51%), or specialist assessment (28%), compared with 37% in 2002 (RR 2.24 [2.15, 2.34]) (Table 2).

Treatment initiation and dosages over time. Amongst patients with a diagnosis of heart failure recorded in primary care, reduced ejection fraction, and no contraindications or intolerances, prescriptions of essential drugs within three months of incident diagnosis differed by drug class. In 2014, prescription rates were high for ACE-Is or ARBs (80%) and beta blockers (72%) but lower for MRAs (28%) (Table 2).

Average daily doses prescribed over the 12 months following diagnosis remained below the recommended targets doses throughout the study period (2014 values: 42% for ACE-Is or ARBs, 29% for beta blockers, 22% for MRAs). When taken combined, the overall dose of beta blockers and ACE-Is/ARBs prescribed to patients changed little over time (adjusted difference from 2002 to 2014: +6% [+2%, +10%]), although patterns differed by individual drug classes and declined for ACE-Is or ARBs (adjusted difference from 2002 to 2014: -7% [-13%, -2%]) (Table 2).

Table 1. Baseline characteristics of patients with incident HF by diagnosis care setting.

Characteristic	All Patients		Diagnosis Care Setting	
	(<i>n</i> = 93,074)	Inpatient (HF Primary Cause) (<i>n</i> = 11,578, 12%)	Inpatient (HF Secondary Cause) (n = 40,789, 44%)	Outpatient (Specialist or Primary Care) (n = 40,707, 44%)
Age [years], mean (SD)	76.7 (12.6)	78.1 (12.6)	77.6 (12.4)	75.4 (12.7)
Women, no. (%)	45,647 (49%)	6,054 (52%)	20,844 (51%)	18,749 (46%)
Ethnicity, no. (%)				
White	56,011 (88%)	6,974 (88%)	25,259 (88%)	23,778 (87%)
Missing	29,122 (31%)	3,609 (31%)	12,132 (30%)	13,381 (33%)
Socioeconomic status, no. (%)				
1 (least deprived)	18,371 (20%)	2,182 (19%)	7,889 (19%)	8,300 (20%)
2	20,073 (22%)	2,492 (21%)	8,600 (21%)	8,981 (22%)
3	20,052 (22%)	2,479 (21%)	8,697 (21%)	8,876 (22%)
4	18,308 (20%)	2,307 (20%)	8,122 (20%)	7,879 (19%)
5 (most deprived)	16,270 (17%)	2,118 (18%)	7,481 (18%)	6,671 (16%)
Systolic blood pressure				
Mean (SD) [mmHg]	133 (21)	133 (21)	132 (20)	133 (201)
Missing, no. (%)	5,195 (6%)	654 (6%)	2,656 (7%)	1,885 (5%)
Diastolic blood pressure				
Mean (SD) [mmHg]	74 (12)	74 (12)	74 (11)	75 (12)
Missing, no. (%)	5,195 (6%)	654 (6%)	2,656 (7%)	1,885 (5%)
BMI category, no. (%)				
Underweight	2,193 (4%)	284 (4%)	1,107 (5%)	802 (3%)
Normal	17,381 (31%)	2,164 (33%)	7,657 (32%)	7,560 (30%)
Overweight	18,786 (34%)	2,067 (31%)	7,818 (33%)	8,901 (35%)
Obese	17,644 (31%)	2,139 (32%)	7,162 (30%)	8,343 (33%)
Missing	37,070 (40%)	4,924 (43%)	17,045 (42%)	15,101 (37%)
Smoking, no. (%)				
No	29,551 (41%)	3,787 (44%)	12,647 (40%)	13,117 (41%)
Ex	32,572 (45%)	3,683 (43%)	14,299 (46%)	14,590 (46%)
Yes	9,596 (13%)	1,082 (13%)	4,360 (14%)	4,154 (13%)
Missing	21,355 (23%)	3,026 (26%)	9,483 (23%)	8,846 (22%)
Comorbidities				
Atrial fibrillation, no. (%)	36,950 (40%)	5,199 (45%)	17,325 (42%)	14,426 (35%)
Chronic kidney disease, no. (%)	22,762 (24%)	3,233 (28%)	11,336 (28%)	8,193 (20%)
Chronic obstructive pulmonary disease, no. (%)	17,896 (19%)	2,113 (18%)	9,276 (23%)	6,507 (16%)
Diabetes, no. (%)	20,531 (22%)	2,952 (25%)	9,412 (23%)	8,167 (20%)
Dyslipidaemia, no. (%)	25,958 (28%)	3,281 (28%)	12,524 (31%)	10,153 (25%)
Hypertension, no. (%)	62,419 (67%)	8,226 (71%)	28,776 (71%)	25,417 (62%)
Ischaemic heart disease, no. (%)	45,584 (49%)	5,804 (50%)	22,247 (55%)	17,533 (43%)
Osteoarthritis, no. (%)	40,176 (43%)	5,029 (43%)	18,227 (45%)	16,920 (42%)
3 or more comorbidities, no. (%)	73,610 (79%)	9,567 (83%)	34,904 (86%)	29,139 (72%)

Number and percentage of records with missing data are displayed for variables with missing entries. Category percentages refer to complete cases. Socioeconomic status refers to IMD 2015 quintile, with 1 referring to the most affluent and 5 to the most deprived quintile. Number of comorbidities refers to any of the 17 conditions investigated (see <u>Methods</u>). **Abbreviations**: BMI, body-mass index; HF, heart failure; IMD, Index of Multiple Deprivation.

https://doi.org/10.1371/journal.pmed.1002805.t001

Fig 2 provides a graphical overview of care delivery indicators and diverging temporal trends.

	Denominator Cohort (n)	2002-2014	2002	2014		p for Trend
Diagnosis Care Setting	All patients with HF (93,074)	n (%)	n (%)	n (%)	RR [95% CI]	
Inpatient (HF primary cause)		11,578 (12%)	912 (12%)	740 (12%)	0.94 [0.86, 1.03]	0.02
Inpatient (HF secondary cause)		40,789 (44%)	2,277 (31%)	3,242 (52%)	1.67 [1.6, 1.74]	< 0.001
Outpatient (specialist or primary care)		40,707 (44%)	4,120 (56%)	2,260 (36%)	0.64 [0.62, 0.67]	< 0.001
Posthospitalisation Follow- Up		n (%)	n (%)	n (%)	RR [95% CI]	
All inpatients	Inpatients, discharged alive (40,856)	6,807 (17%)	489 (20%)	459 (14%)	0.73 [0.65, 0.82]	< 0.001
Inpatient (HF primary cause)	Primary inpatients, discharged alive (9,195)	2,854 (31%)	228 (32%)	174 (29%)	0.95 [0.80, 1.12]	0.209
Inpatient (HF secondary cause)	Secondary inpatients, discharged alive (31,661)	3,953 (12%)	261 (15%)	285 (11%)	0.73 [0.62, 0.85]	< 0.001
Diagnostic Investigations	Patients diagnosed or followed up in primary care (47,925)	n (%)	n (%)	n (%)	RR [95% CI]	
Echocardiogram		24,649 (51%)	796 (17%)	1,693 (62%)	3.56 [3.36, 3.78]	< 0.001
ECG		17,928 (37%)	995 (21%)	1,090 (40%)	1.83 [1.71, 1.96]	< 0.001
NP test		4,177 (9%)	NA	616 (23%)	NA	NA
Specialist assessment		14,046 (29%)	563 (12%)	847 (31%)	2.5 [2.27, 2.75]	< 0.001
At least 1 diagnostic investigation		33,660 (70%)	1,683 (37%)	2,248 (82%)	2.24 [2.15, 2.34]	< 0.001
Treatment Initiation	Patients diagnosed or followed up in primary care, reduced ejection fraction (11,040), and no drug-specific contraindication	n (%)	n (%)	n (%)	RR [95% CI]	
ACE-I/ARB	9,920	7,748 (78%)	285 (75%)	586 (80%)	1.06 [0.99, 1.13]	< 0.001
Beta blocker	8,430	4,661 (55%)	72 (22%)	490 (72%)	3.27 [2.65, 4.03]	< 0.001
MRA	10,896	1,999 (18%)	59 (14%)	207 (26%)	1.88 [1.45, 2.44]	< 0.001
Beta blocker and ACE-I/ARB	7,768	3,759 (48%)	56 (18%)	403 (63%)	3.48 [2.72, 4.43]	< 0.001
Treatment Dose	Patients diagnosed or followed up in primary care, reduced ejection fraction (11,040) and no drug-specific contraindication	mean (SD)	mean (SD)	mean (SD)	Adjusted difference [95% CI]	
ACE-I/ARB	9,920	48% (45%)	50% (44%)	42% (41%)	-7% [-13%, -2%]	< 0.001
Beta blocker	8,430	25% (30%)	11% (22%)	29% (32%)	19% [15%, 22%]	< 0.001
MRA	10,896	18% (39%)	16% (36%)	20% (37%)	5% [1, 9%]	< 0.001
Beta blocker and ACE-I/ARB	7,768	36% (30%)	30% (26%)	35% (29%)	6% [2%, 10%]	< 0.001

Table 2. Temporal trends in essential care indicators following incident HF by year of diagnosis.

RRs or adjusted differences and 95% CIs comparing 2014 to 2002, adjusting for year of diagnosis, age, sex, socioeconomic status, and region. Primary care follow-up refers to the documentation of HF in primary care records during a follow-up consultation within 12 months of an incident diagnosis in hospital. Diagnostic investigations refer to investigations referred for within ±6 months of incident heart failure. Treatment initiation refers to the issue of at least 1 prescription within 3 months of incident heart failure. Treatment dose presents the average daily dose prescribed in the first 12 months following incident heart failure as percent of guideline-recommended target dose. **Abbreviations**: ACE-I, angiotensin-converting–enzyme inhibitor; ARB, angiotensin receptor blocker; CI, confidence interval; ECG, electrocardiogram; HF, heart failure; MRA, mineralocorticoid receptor antagonist; NA, not applicable; NP, natriuretic peptide; RR, risk ratio.

https://doi.org/10.1371/journal.pmed.1002805.t002

Treatment titration patterns before and after incident diagnosis

Prescription rates of the investigated drug classes in the three months preceding heart failure diagnosis were substantial (44% for ACE-Is or ARBs, 22% for beta blockers, and 3% for MRAs) and were largely related to an existing diagnosis of hypertension (RR for the prescription of at least one treatment class in patients with hypertension versus no hypertension: 2.07 [1.96, 2.18]).

Dose increments happened largely in the first 30 days following diagnosis, with no evidence of consistent increments thereafter (Fig 3). Because estimation of average doses across the whole population could potentially mask changes in dosing at the individual patient level, we further investigated changes to treatment dose by categories of 'no change', 'increase', or 'decrease' at fixed time intervals of 2 and 12 months after diagnosis. This showed that even in more recent years (2012–2014), for about half of the patients (52%), there was no change in dosage of ACE-Is or ARBs between month 2 and month 12 after diagnosis. The treatment dose decreased in 19% and increased in 29% of patients. At the end of one year, 27% of patients received no ACE-I or ARB treatment, and another 36% were on <50% of the guideline-



Fig 2. Care delivery indicators following incident heart failure, by year of diagnosis, in CPRD from 2002 to 2014. Results are presented as fitted local polynomial regression over yearly averages and 95% CIs (dashed lines). ACE-I, angiotensin-converting–enzyme inhibitor; ARB, angiotensin receptor blocker; CI, confidence interval; CPRD, Clinical Practice Research Datalink; ECG, electrocardiogram; NP, natriuretic peptide.

https://doi.org/10.1371/journal.pmed.1002805.g002

PLOS MEDICINE

recommended dose (**Fig 4A**). The patterns of changes were similar for beta blockers, for which a year after diagnosis, 30% did not receive this treatment (**Fig 4B**). Although overall doses changed over time, patterns of up-titration remained similar across time periods (**S1 Fig**).

To allow comparison with clinical trials and cross-sectional observational studies, we further report supplementary analyses on maximal prescribed doses as well as dosages amongst patients initiated on therapy (S7 Table).

Stratified analyses by age, sex, and socioeconomic status. We observed variations in care by age, sex, and socioeconomic status (Fig 5).

Women were less commonly diagnosed in outpatient settings and had lower rates of primary care follow-up, fewer diagnostic investigations, less treatment initiation, and lower treatment doses than their male counterparts of similar age, socioeconomic status, and region.

Older patients (aged \geq 75 years) exhibited similar and significant differences compared with their middle-aged counterparts (55 to 64 years old). The very young (<45 years) patients, particularly young women, had lower rates of diagnostic investigations across all four test types compared with the 55 to 64 years age group (S8 Table).

Amongst patients managed by their general practitioner, no socioeconomic disparities were apparent with regard to prescriptions of treatment or dose of treatment. However, deprived patients had lower rates of diagnosis in outpatient settings than their more affluent counterparts and hence were overall less likely to receive primary care follow-up.



Fig 3. Average daily dose of guideline-recommended treatments prescribed around the time of incident heart failure in patients diagnosed from 2012–2014. Average daily dose prescribed to patients with heart failure and reduced ejection fraction without drug-specific contraindications or intolerances, from 3 months prior up to 12 months following incident heart failure, in patients diagnosed between 2012 and 2014, smoothed with local polynomial regression. Average daily dose is expressed as a percentage of the guideline-recommended target dose. ACE-I, angiotensin-converting-enzyme inhibitor; ARB, angiotensin receptor blocker; MRA, mineralocorticoid receptor antagonist.

https://doi.org/10.1371/journal.pmed.1002805.g003

Disparities remained apparent, although attenuated after adjusting for differences in follow-up times, 30-day mortality, and baseline comorbidities (S2 Text), with the exception of differences in primary care follow-up in patients aged \geq 75 years.

Discussion

This large-scale, population-based study provides important information on contemporary care of heart failure patients in routine clinical practice and insights into its variation over time by age, sex, and socioeconomic status. Our study confirms previous reports of high rates of guideline-indicated diagnostic investigations and treatment initiation in Western countries (S1 Table). However, further investigation of care across the continuum of primary and secondary services and from the prediagnosis stage to several months after incident diagnosis revealed important shortcomings in the management of patients. First, rates of outpatient diagnoses and follow-up in primary care after hospital discharge are low and have been declining over time. Second, doses of key medicines remain far below those recommended in guide-lines in all groups of patients and for all three drug classes investigated, even a year after diagnosis. Finally, deficiencies in care were more common in women, older people, and, to some extent, socioeconomically deprived individuals.

Indeed, we found that rates of outpatient diagnoses have been declining whilst more patients are being diagnosed acutely in hospital. Relatedly, we found that only 17% of patients who were first diagnosed in hospital were subsequently followed up with a heart failure



Fig 4. Drug dose trajectory of (A) ACE-I or ARB dose and (B) beta blocker, prescribed around the time of incident heart failure, in patients diagnosed from 2012–2014. Drug dose trajectory prescribed to patients with heart failure and reduced ejection fraction without contraindications or intolerances, in patients with incident heart failure between 2012 and 2014. 'Month –1' presents the average daily dose prescribed in the 30 days preceding incident heart failure. 'Month 2' presents the average daily dose prescribed in the second month (days 30 to 60) following incident heart failure. 'Month 12' presents the average daily dose prescribed in the twelfth month (days 335 to 365) following incident heart failure. ACE-I, angiotensin-converting–enzyme inhibitor; ARB, angiotensin receptor blocker.

https://doi.org/10.1371/journal.pmed.1002805.g004

diagnosis in primary care, and this rate has also been declining, in particular when heart failure was not the primary discharge diagnosis. The reasons for the declining trends in diagnosis and follow-up of heart failure patients outside the acute hospital setting are not entirely clear and seem rather surprising when put in the context of the growing access to diagnostic services such as blood BNP outside hospitals and the reported shift in diagnosis from hospitals to outpatient settings elsewhere [21]. Our findings suggest that out-of-hospital screening and followup are suboptimal, and this could, at least partly, be due to poor record-keeping in primary care and inadequate information exchange between hospitals and primary care. It might also be that the national primary care reporting and incentive scheme [22] itself paradoxically contributed to these trends, with general practitioners, for instance, using free text descriptions to record patients' problems rather than formally recording heart failure as a diagnosis. Such practices that avoid registering certain patients who will not achieve management recommendations to achieve higher overall adherence rates have been reported for other chronic conditions, such as depression [23,24]. Recent investments in specialist heart failure clinics and nurse-led services [25] might also have contributed to the observed patterns by creating artificially low primary care diagnosis and follow-up rates. Support for this comes from our sensitivity analyses, which indicate that some patients receive heart failure medicines in primary care despite no formal documentation of heart failure (S1 Text).

However, irrespective of the underlying reasons, the declining rates of heart failure recording in primary care are likely to have important consequences for patient care as well as research. The UK healthcare system relies on primary care as the cornerstone of chronic disease management. In such systems, general practitioners take a central role in screening, coordination, and continuous management of common conditions such as heart failure. Even

no	Diagnosis in tpatient setting	Post-hospital follow-u	lisation Ip	Diagnost investigati	ic ons	Treatmer initiation	ŧ	Treat	tment se	
Age group < 45 45-54 55-64 65-74 65-74	Hendrich 1.05 [0.39, 1.10] Hendrich 1.01 [0.97, 1.05] Hendrich 1.00 [Reference] 0.96 [0.34, 0.39] 0.05 [0.33, 0.87]	II.I	0.96 [0.82, 1.13] 0.98 [0.87, 1.11] 1.00 [Reference] 0.93 [0.86, 1.01] 0.89 [0.82, 0.95]		0.84 [0.81, 0.88] 0.97 [0.94, 1.00] 1.00 [Reference] 1.00 [0.98, 1.02] 0.88 [0.86, 0.90]	I I	0.89[0.80, 0.99] 1.02[0.94, 1.10] 1.00[Reference] 0.96[0.90, 1.03] 0.80[0.75, 0.85]		1 [0.78 1.1 [0.9 1.0 [Refe 0.84 [0.74] 0.52 [0.43]	, 1.29] , 1.34] rence] , 1.00]
Sex Men Women	1.00 [Reference]	•	1.00 [Reference] 0.82 [0.79, 0.86]	•	1.00 [Reference] 0.92 [0.91, 0.93]	•	1.00 [Reference] 0.87 [0.83, 0.91]	I	1.00 [Refe 0.78 [0.68	irence] , 0.90]
Socioeconomic quintile 1 (least deprived) 2 3 4 5 (most deprived)	 1.00 [Reference] 0.99 [0.97, 1.01] 0.98 [0.96, 1.00] 0.94 [0.92, 0.96] 0.98 [0.36, 0.30] 	• <u>1</u> <u>1</u> <u>1</u>	1.00 [Reference] 1.04 [0.97, 1.12] 1.04 [0.97, 1.11] 1.06 [0.99, 1.14] 1.00 [0.93, 1.07]		1.00 [Reference] 1.00 [0.98, 1.01] 0.96, 0.99] 0.95 [0.33, 0.97] 0.95 [0.33, 0.97]	• • • • •	1.00 [Reference] 1.01 [0.94, 1.09] 0.99 [0.92, 1.06] 0.98 [0.91, 1.06] 1.04 [0.96, 1.12]		1.00 [Refe 1.15 [0.95 0.94 [0.738	rence] 1.1.40] 1.1.44] 1.1.16]
0.5 0.7	5 1 1.25 0.5	0.75	۲ 1.25 0.5	0.75 1	ے 1.25 0.5		0 72	5 0.75 1	1.25	
Fig 5. Stratified analysis of Poisson regression model ad assessment, within ± 6 month attheory recorded contribution	care delivery indicators by i justed for year of diagnosis, i as of incident heart failure dir cations or involvements. Treat	ige, sex, and soci ge group, sex, soc ignosis. Treatmer	oeconomic quintile cioeconomic status, a nts refer to the prescr odicates at least 1 mm	in patients wi and region. Di ription of both	th heart failure in C agnostic investigatio (i) ACE-Is or ARBs in 3 months of incirc	PRD from 200 ns refer to eithe and (ii) BBs in	2 to 2014. RRs and r one of echocardic patients with heart	l 95% CIs fron ogram, ECG, N failure and rev	n multivariable NPs, or specialist duced ejection fi	action



where specialist clinics and nurse-led community services exist, general practitioners remain responsible for medication prescriptions and the coordination of specialty care. Accurate disease recording in primary care is therefore particularly important for this patient population with a high number of both cardiovascular and noncardiovascular comorbidities [15], and gaps in healthcare records are likely to have effects on subsequent monitoring and management of patients. This under-registration might also explain why some previous analyses concluded that the incidence of heart failure is declining in the UK [26] (which was refuted in subsequent studies that used linked records to capture cases across primary and secondary care) [15].

The increase in the proportion of patients initiated on evidence-based therapy for heart failure in recent years is important and encouraging because prescription of treatment, even at low dose, is an essential component of heart failure management. Yet gaps remain, particularly for beta blockers and MRAs. Whilst comparison across studies is difficult because of varying inclusion criteria and time points at which treatment plans are evaluated, rates appear somewhat lower than those reported in other countries (S1 Table), and underuse of these therapies are likely to be associated with significant loss of life, both quantity and quality [27]. Moreover, the average dose received per patient remained far below the guideline-recommended doses, and despite increasing numbers of patients initiated on therapy, the overall dose prescribed did not change substantially from 2002 to 2014. Titration was largely restricted to the 30-day period after incident diagnosis, yet landmark clinical trials have established the value of optimal drug dosages in conferring clinical benefits [28-30]. Whilst it may not be possible for every patient to achieve guideline-recommended doses, the rates reported by our study remain below rates achieved through community-based interventions or in clinical trials [31]. Our findings hence question whether current efforts in community heart failure services sufficiently address the long-term needs of patients with heart failure. In addition, they show that the current UK focus on monitoring diagnostic tests and treatment initiation might distract attention away from medication maintenance and titration as a measure of quality of care.

Our study further shows that the management of women, older people, and deprived individuals was even less satisfactory. For example, all were more likely to be first diagnosed during a hospital admission. This may suggest that, in these patients, early signs and symptoms have not been appropriately recognised in nonacute healthcare settings. Alternatively, this may be related to patient preferences to seek care in hospital or, for deprived populations, to more difficult access to outpatient consultations [32]. Correcting such disparities is an important challenge for a system that intends to offer equality of access to and quality of care.

Healthcare reporting efforts play an important role in improving quality of care. Our study presents changes in the delivery of care alongside the implementation of landmark quality improvement initiatives, in particular the QOF, a reporting and incentives scheme for primary care introduced in 2004 [22], and the NHFA, a reporting programme for secondary care introduced in 2007 [11]. These schemes represent the primary source of quality of care data in the UK so far and report a very positive picture of heart failure management. For example, both report very high rates of diagnostic investigations (91%–95%) [12,13] and treatment initiation (85%–99% for ACE-Is or ARBs) [12,13] that, given the small differences in variable definitions, are comparable to the present findings. However, the investigation of care across the continuum of primary and secondary services reveals important shortcomings in the management of patients that analyses confined to individual clinical settings are unable to identify. When reflecting specifically on the financial incentives programme, which applies to primary care, it appears that its design insufficiently addresses patients' long-term care needs. The scheme incentivises general practitioners to follow evidence-based care practices and monitors over 100 indicators across a range of conditions. For heart failure, the scheme monitors referral for

diagnostic investigations and yearly prescriptions of both beta blockers and ACE-Is or ARBs, with financial rewards only for primary care practices that achieve set target rates. Our findings reveal that incentivised indicators have improved over time but show no or negative changes in associated outcomes. Perhaps most importantly, they reveal possible unintended consequences in the fact that following the introduction of the programme, recording of diagnoses in primary care declined considerably, leading to a large and growing number of patients for whom continuous monitoring and management remains uncertain.

A major strength of this study is the use of a large population-based cohort of patients, which allows a sufficient number of cases in each age, sex, and socioeconomic category for subpopulation analyses and increases the generalisability of findings compared with surveys enrolling more selected participants. Moreover, longitudinal data from electronic health records provide a unique opportunity to follow patients over time in different care settings and hence to address the limitations of selected single-setting and single-time-point indicators used in previous studies. One of the key limitations of our study was the incomplete clinical information contained in the available electronic health records. In particular, left ventricular ejection fraction values were not available. Hence, we could only characterise the type of heart failure in a subset of patients for whom the diagnostic code made a clear reference to reduced ejection fraction. Whilst this subset of patients is likely to underestimate the true prevalence of reduced ejection fraction in the community, its high specificity ensures the selection of patients is appropriate for our analyses. Moreover, secondary care records did not provide access to diagnostic investigations, procedures (such as device implantations), discharge prescriptions or referrals, or outpatient consultations, so some analyses had to be restricted to primary care data. Finally, reliable information about patients' symptoms was not available and limited our ability to investigate precise indications for certain therapies, such as MRAs.

Our findings have important implications for health services policies. The increased uptake of guideline-recommended diagnostic tests and treatment initiation amongst patients seen in primary care suggests that early management of these patients has improved, probably because of a combination of physician awareness, clinical guidelines, and financial incentives. However, the limited changes to medication dosages, the disparities amongst subgroups of patients, and the poor rates of primary care recording indicate that more efforts are needed. Quality improvement efforts that remained confined within individual care settings have proven insufficient to identify important care gaps and to address challenges of this chronic condition with effective but complex treatment. Further improvements are likely to require a broader perspective to health services design to support appropriate care at every level of the patient journey.

Supporting information

S1 Text. Documentation of diagnosis in primary care records as a prerequisite for subsequent disease monitoring and management. (DOCX)

S2 Text. Method for the calculation of average daily doses. (DOCX)

S3 Text. Sensitivity analyses. (DOCX)

S4 Text. RECORD statement. RECORD, REporting of studies Conducted using Observational Routinely collected health Data. (DOCX) **S5 Text. Study design and analysis plan.** (DOCX)

S1 Table. Selected studies reporting care delivery in patients with heart failure. (DOCX)

S2 Table. Clinical codes used to identify patients with heart failure and reduced ejection fraction from general practice records. (DOCX)

S3 Table. Clinical codes used to identify drug-class-specific contraindications or intolerances from general practice records. (DOCX)

S4 Table. Clinical codes used to identify diagnostic tests performed in patients with heart failure from general practice records.

(DOCX)

S5 Table. Recommended maintenance dosages of pharmacological treatments indicated in patients with heart failure and reduced ejection fraction, as per ESC and NICE guidelines during the study period. ESC, European Society of Cardiology; NICE, National Institute for Clinical Excellence.

(DOCX)

S6 Table. Baseline characteristics of patients with incident heart failure by record of ejection fraction.

(DOCX)

S7 Table. Temporal trends in supplementary care indicators following incident heart failure by year of diagnosis. (DOCX)

S8 Table. Diagnostic investigations following incident heart failure, stratified by age and sex.

(DOCX)

S1 Fig. Average daily dose of guideline-recommended treatments prescribed around the time of incident heart failure by time period of diagnosis. (DOCX)

Author Contributions

Conceptualization: Nathalie Conrad, John J. V. McMurray, Kazem Rahimi.

Data curation: Nathalie Conrad.

Formal analysis: Nathalie Conrad, Andrew Judge.

Funding acquisition: Nathalie Conrad, Kazem Rahimi.

- Investigation: Nathalie Conrad, Andrew Judge, Dexter Canoy, Jenny Tran, Johanna O'Donnell, F. D. Richard Hobbs, John G. Cleland, John J. V. McMurray, Kazem Rahimi.
- Methodology: Nathalie Conrad, Andrew Judge, Dexter Canoy, Jenny Tran, Johanna O'Donnell, Milad Nazarzadeh, Gholamreza Salimi-Khorshidi, F. D. Richard Hobbs, John G. Cleland, John J. V. McMurray, Kazem Rahimi.

Project administration: Nathalie Conrad.

Resources: Kazem Rahimi.

Supervision: Andrew Judge, Dexter Canoy, Kazem Rahimi.

Visualization: Nathalie Conrad.

- Writing original draft: Nathalie Conrad, Kazem Rahimi.
- Writing review & editing: Nathalie Conrad, Andrew Judge, Dexter Canoy, Jenny Tran, Johanna O'Donnell, Milad Nazarzadeh, Gholamreza Salimi-Khorshidi, F. D. Richard Hobbs, John G. Cleland, John J. V. McMurray, Kazem Rahimi.

References

- 1. National Institute for Clinical Excellence (NICE). Management of chronic heart failure in adults in primary and secondary care. Clinical Guideline 5. London: Royal College of Physicians (UK); 2003.
- National Institute for Health and Clinical Excellence (NICE). Chronic heart failure: management of chronic heart failure in adults in primary and secondary care (CG108) [Internet] [cited 2018 June 4].
 2010. Available from: http://guidance.nice.org.uk/CG108/Guidance
- Remme WJ, Swedberg K. Task Force Report Guidelines for the diagnosis and treatment of chronic heart failure Diagnosis of chronic heart failure. Eur Heart J. 2001; 22: 1527–1560. https://doi.org/10. 1053/euhj.2001.2783 PMID: 11492984
- 4. Swedberg K, Cleland J, Dargie H, Drexler H, Follath F, Komajda M, et al. Guidelines for the diagnosis and treatment of chronic heart failure: executive summary (update 2005): The Task Force for the Diagnosis and Treatment of Chronic Heart Failure of the European Society of Cardiology. Eur Heart J. Oxford University Press; 2005; 26: 1115–1140. https://doi.org/10.1093/eurheartj/ehi204 PMID: 15901669
- Dickstein K, Cohen-Solal A, Filippatos G, McMurray JJ, Ponikowski P, Poole-Wilson PA, et al. ESC guidelines for the diagnosis and treatment of acute and chronic heart failure 2008: the Task Force for the diagnosis and treatment of acute and chronic heart failure 2008 of the European Society of Cardiology. Developed in collaboration with the Heart. Eur J Hear Fail. 2008/10/02. 2008; 10: 933–989. doi: S1388-9842(08)00370-X [pii] https://doi.org/10.1016/j.ejheart.2008.08.005 PMID: 18826876
- McMurray JJ V, Adamopoulos S, Anker SD, Auricchio A, Böhm M, Dickstein K, et al. ESC Guidelines for the diagnosis and treatment of acute and chronic heart failure 2012: The Task Force for the Diagnosis and Treatment of Acute and Chronic Heart Failure 2012 of the European Society of Cardiology. Developed in collaboration with the Heart. Eur Heart J. 2012; 33: 1787–847. https://doi.org/10.1093/ eurhearti/ehs104 PMID: 22611136
- Ponikowski P, Voors AA, Anker SD, Bueno H, Cleland JGF, Coats AJS, et al. 2016 ESC Guidelines for the diagnosis and treatment of acute and chronic heart failure. Eur Heart J. 2016; 37: 2129–2200. https://doi.org/10.1093/eurheartj/ehw128 PMID: 27206819
- Komajda M, Anker SD, Cowie MR, Filippatos GS, Mengelle B, Ponikowski P, et al. Physicians' adherence to guideline-recommended medications in heart failure with reduced ejection fraction: data from the QUALIFY global survey. Eur J Heart Fail. John Wiley & Sons, Ltd; 2016; 18: 514–522. <u>https://doi.org/10.1002/ejhf.510</u> PMID: 27095461
- Callender T, Woodward M, Roth G, Farzadfar F, Lemarie J-C, Gicquel S, et al. Heart failure care in lowand middle-income countries: a systematic review and meta-analysis. PLoS Med. Public Library of Science; 2014; 11: e1001699. https://doi.org/10.1371/journal.pmed.1001699 PMID: 25117081
- The Health and Social Care Information Centre. The Quality and Outcomes Framework (QOF) 2004/05 Background [Internet]. 2012 [cited 18 Jul 2018]. Available from: https://data.england.nhs.uk/dataset/ qof-national-quality-outcomes-framework-2004-05/resource/3c904948-ac3f-4faa-b118-615e8282eee26
- National Institute for Cardiovascular Outcomes Research (NICOR) UCL. National Heart Failure Audit. April 2015-March 2016. [Internet]. 2017 [cited 2018 June 4]. Available from: https://www.nicor.org.uk/ wp-content/uploads/2019/02/annual-report-2015-6-v8.pdf.
- Health and Social Care Information Centre. Prevalence, Achievements and Exceptions Report from the Quality and Outcomes Framework, England 2013–14 [Internet]. 2014 [cited 2018 June 6]. Available from: http://content.digital.nhs.uk/catalogue/PUB15751/qof-1314-report-V1.1.pdf
- Mitchell P, Marle D, Donkor A, Shote A, McDonagh T, Hardman S, et al. National Heart Failure Audit 2013–2014 [Internet]. 2015 [cited 2018/06/04]. Available from: https://www.nicor.org.uk/wp-content/ uploads/2019/02/hfannual13-14-updated.pdf

- Herrett E, Gallagher AM, Bhaskaran K, Forbes H, Mathur R, van Staa T, et al. Data Resource Profile: Clinical Practice Research Datalink (CPRD). Int J Epidemiol. Oxford University Press; 2015; 44: 827– 36. https://doi.org/10.1093/ije/dyv098 PMID: 26050254
- Conrad N, Judge A, Tran J, Mohseni H, Hedgecott D, Crespillo AP, et al. Temporal trends and patterns in heart failure incidence: a population-based study of 4 million individuals. Lancet. Elsevier; 2017; 391: 572–580. https://doi.org/10.1016/S0140-6736(17)32520-5 PMID: 29174292
- CALIBER Research Portal [Internet]. [cited 1 Feb 2017]. Available from: https://www.caliberresearch. org/portal/
- Department for Communities and Local Government (DCLG). The English Index of Multiple Deprivation 2015: Guidance [Internet]. 2015 [cited 2017 Feb 2]. Available from: https://www.gov.uk/government/statistics/english-indices-of-deprivation-2015
- 18. Loess function | R Documentation [Internet]. [cited 1 Feb 2019]. Available from: https://www. rdocumentation.org/packages/stats/versions/3.5.2/topics/loess
- Cleveland WS, Devlin SJ. Locally Weighted Regression: An Approach to Regression Analysis by Local Fitting. J Am Stat Assoc. 1988; 83: 596–610. https://doi.org/10.1080/01621459.1988.10478639
- Benchimol EI, Smeeth L, Guttmann A, Harron K, Moher D, Petersen I, et al. The REporting of studies Conducted using Observational Routinely-collected health Data (RECORD) Statement. PLoS Med. 2015; 12: e1001885. https://doi.org/10.1371/journal.pmed.1001885 PMID: 26440803
- Ezekowitz JA, Kaul P, Bakal JA, Quan H, McAlister FA. Trends in heart failure care: has the incident diagnosis of heart failure shifted from the hospital to the emergency department and outpatient clinics? Eur J Heart Fail. 2011; 13: 142–147. https://doi.org/10.1093/eurjhf/hfg185 PMID: 20959343
- Doran T, Fullwood C, Gravelle H, Reeves D, Kontopantelis E, Hiroeh U, et al. Pay-for-Performance Programs in Family Practices in the United Kingdom. N Engl J Med. Massachusetts Medical Society; 2006; 355: 375–384. https://doi.org/10.1056/NEJMsa055505 PMID: 16870916
- Kendrick T, Stuart B, Newell C, Geraghty AWA, Moore M. Changes in rates of recorded depression in English primary care 2003–2013: Time trend analyses of effects of the economic recession, and the GP contract quality outcomes framework (QOF). J Affect Disord. 2015; 180: 68–78. <u>https://doi.org/10.1016/j.jad.2015.03.040</u> PMID: 25881283
- Roland M, Guthrie B. Quality and Outcomes Framework: what have we learnt? BMJ. 2016; 354:i4060. https://doi.org/10.1136/bmj.i4060 PMID: 27492602
- Grange J. The role of nurses in the management of heart failure. Heart. BMJ Publishing Group Ltd; 2005; 91 Suppl 2: ii39–42; discussion ii43–8. https://doi.org/10.1136/hrt.2005.062117 PMID: 15831611
- Hawkins NM, Scholes S, Bajekal M, Love H, O'Flaherty M, Raine R, et al. Community care in England: reducing socioeconomic inequalities in heart failure. Circulation. 2012/07/28. 2012; 126: 1050–1057. https://doi.org/10.1161/CIRCULATIONAHA.111.088047 PMID: 22837162
- Walker S, Spackman E, Conrad N, Emdin CA, Griffin E, Rahimi K, et al. Impact of missed treatment opportunities on outcomes in hospitalised patients with heart failure. Open Hear. Archives of Disease in childhood; 2017; 4: e000726. https://doi.org/10.1136/openhrt-2017-000726 PMID: 29296284
- Packer M, Poole-Wilson PA, Armstrong PW, Cleland JGF, Horowitz JD, Massie BM, et al. Comparative Effects of Low and High Doses of the Angiotensin-Converting Enzyme Inhibitor, Lisinopril, on Morbidity and Mortality in Chronic Heart Failure. Circulation. American Heart Association, Inc.; 1999; 100: 2312– 2318. https://doi.org/10.1161/01.cir.90.5.2457 PMID: 10587334
- Konstam MA, Neaton JD, Dickstein K, Drexler H, Komajda M, Martinez FA, et al. Effects of high-dose versus low-dose losartan on clinical outcomes in patients with heart failure (HEAAL study): a randomised, double-blind trial. Lancet. Elsevier; 2009; 374: 1840–1848. <u>https://doi.org/10.1016/S0140-6736</u> (09)61913-9 PMID: 19922995
- Bristow MR, Gilbert EM, Abraham WT, Adams KF, Fowler MB, Hershberger RE, et al. Carvedilol Produces Dose-Related Improvements in Left Ventricular Function and Survival in Subjects With Chronic Heart Failure. Circulation. 1996; 94: 2807–2816. Available from: <u>http://circ.ahajournals.org/cgi/content/</u> abstract/94/11/2807 PMID: 8941106
- **31.** Lowrie R, Mair FS, Greenlaw N, Forsyth P, McConnachie A, Richardson J, et al. The Heart failure and Optimal Outcomes from Pharmacy Study (HOOPS): Rationale, design, and baseline characteristics. Eur J Heart Fail. 2011; 13: 917–924. https://doi.org/10.1093/eurjhf/hfr083 PMID: 21791543
- 32. McAlister FA, Murphy NF, Simpson CR, Stewart S, MacIntyre K, Kirkpatrick M, et al. Influence of socioeconomic deprivation on the primary care burden and treatment of patients with a diagnosis of heart failure in general practice in Scotland: population based study. BMJ. 2004; 328.